Author's response to reviews

Title: Modulation of T cell proliferation and cytokine response by Plumbagin extracted from Plumbago zeylanica in collagen induced arthritis

Authors:

Aparanji Poosarla (apara_p@yahoo.co.in)
Rao D N (dnrao@hotmail.com)
Rama Rao Athota (raaathota@yahoo.com)
Venu Gopal Sunkara (venugsunkara@gmail.com)

Version: 8 Date: 8 November 2011

Author's response to reviews: see over
Dear Sir,

We have incorporated all the corrections in the manuscript as indicated in your letter.

1. E-mail addresses of all the authors were added in title page
2. Reference numbers in the text were changed according to the format.
3. It was stated that all the authors have read and approved the final manuscript
4. Figures were altered as per the guidelines

We are submitting our manuscript with all above points for your kind perusal

Thanking you

Yours sincerely

Aparanji Poosarla